

Check for updates

# **Evolution of clinical trials in multiple** sclerosis

Yinan Zhang, Amber Salter, Erik Wallström, Gary Cutter and Olaf Stüve

Abstract: Clinical trials have advanced the treatment of multiple sclerosis (MS) by demonstrating the safety and efficacy of disease-modifying therapies (DMTs). This review discusses major changes to MS clinical trials in the era of DMTs. As treatment options for MS continue to increase, patients in modern MS trials present earlier and with milder disease compared with historic MS populations. While placebo-controlled trials for some questions may still be relevant, DMT trials in relapsing-remitting MS (RRMS) are no longer ethical. The replacement of the placebo arm by an active comparator arm in trials have raised the cost of trials by requiring larger sample sizes to detect on-study changes in treatment effects. Efforts to improve trial efficiency in RRMS have focused on exploring adaptive designs and relying on sensitive magnetic resonance imaging measures of disease activity. In trials for progressive forms of MS (PMS), the lack of sensitive outcome measures that can be used in shorter-term trials have delayed the development of effective treatments. Recent shifting of the focus to advancing trials in PMS has identified paraclinical outcome measurements with improved potential, and the testing of agents for neuroprotection and remyelination is in progress.

**Keywords:** multiple sclerosis, clinical trials, trial design, diagnostic criteria, outcome measure, progressive multiple sclerosis

Received: 19 July 2018; revised manuscript accepted: 17 December 2018.

# Introduction

The advancement of new therapeutic agents for multiple sclerosis (MS) relies on well-designed clinical trials to establish their safety and efficacy. Over the last two decades, clinical trials in MS have established a success rate of 27%, defined as passing phase I, II, III and United States Food and Drug Administration (US FDA) approval, almost tripling the overall industry rate of 10%.1 As a result of now having 15 approved diseasemodifying therapies (DMTs) for relapsing-remitting MS (RRMS), there is a greater challenge to improve the existing options and to achieve prolonged remission. The increased availability of treatment along with revisions in the diagnostic criteria have changed the clinical trial population and restricted the implementation of placebo-controlled trials. At the same time, an increased understanding of the pathophysiology and natural history of the disease have spurred the development of new outcome measurements and refinement of existing metrics. Despite numerous successes in advancing therapies for RRMS, similar progress has not been achieved for patients with progressive forms of MS (PMS), and previously failed trials in PMS have delineated the challenges that must be overcome to develop treatments for PMS. The design of MS trials must take its dynamic landscape into consideration and account for the differences between modern and historical trials. This article will review the changing characteristics of MS trial patients, the shift from placebo-controlled trials to active comparator studies, and traditional and new outcome measurements. In addition, the challenges and progress in PMS trials will be discussed.

# Changes in trial populations

The characteristics of RRMS trial populations have changed over time, largely in response to shifting methodologies in patient diagnosis and selection as well as the availability of multiple treatments. Patients in recent trials are presenting with lower disease activity and slower clinical progression.<sup>2</sup> This is evident in the annualized relapse

Ther Adv Neurol Disord

2019, Vol. 12: 1-14

DOI: 10.1177/ 1756286419826547

© The Author(s), 2019. Article reuse guidelines: sagepub.com/journalspermissions

# Correspondence to: Olaf Stüve

Neurology Section, VA North Texas Health Care System, Medical Service, 4500 South Lancaster Rd., Dallas, TX 75216, USA Department of Neurology, Klinikum rechts der Isar, Technische Universität München, Germany

#### olaf.stuve@ utsouthwestern.edu

# Yinan Zhang

Department of Neurology and Neurotherapeutics, the University of Texas Southwestern Medical Center, Dallas, TX, USA

# Amber Salter

Division of Biostatistics, Washington University School of Medicine, St. Louis, MO. USA

# Erik Wallström

Sanofi Genzyme, Neuro and Gene Therapy, Cambridge, MA, USA

#### **Garry Cutter**

Department of Biostatistics, University of Alabama at Birmingham, Birmingham, AL, USA



rates (ARRs) in the treatment groups from historical trials between 1993-2002 that have ranged between 0.5 and 0.87 whereas subsequent modern-day trials have reported ARRs between 0.16 and 0.37.3 While some of the reductions in relapse rates are likely due to the more rigid definitions in trials used to confirm relapses, there nevertheless appears a steady decline in the ARR over time. A report analyzing the ARR in RRMS trials over time demonstrated a reduction of 0.37 relapses in the treatment arms and 0.36 in the placebo arms over a 10-year study period.4 In addition, patients who were relapse-free in the placebo groups of RRMS trials experience a longer time to their first relapse.<sup>2</sup> The trend of MS trial patients presenting with milder disease over time is a result of multiple processes. In this section, we explore the changes in the MS trial population and its implications.

# Changes in diagnostic criteria

The MS diagnostic criteria and its subsequent revisions have made it easier to diagnose the disease and this has resulted in many patients receiving an earlier diagnosis.<sup>5,6</sup> Since the 2010 revision of the McDonald criteria, a diagnosis of MS can be made in a single time frame, in which lesions on a single magnetic resonance imaging (MRI) scan can fulfill both criteria for dissemination in time (DIT) and dissemination in space (DIS). This contrasts with previous criteria in which the diagnosis required subsequent clinical presentations or MRI to fulfill DIT. The latest revision of the McDonald criteria in 2017 further broadened the diagnostic parameters and included the presence of oligoclonal bands in the cerebrospinal fluid to substitute for DIT when the initial MRI findings did not fulfill this requirement. In addition, cortical lesions, and symptomatic lesions excluding the optic nerve, may now be counted as MRI evidence of DIS or DIT. By broadening the diagnostic criteria, more patients are likely to be diagnosed earlier whereas they previously would not be diagnosed by an earlier version of the criteria.

A consequence of earlier diagnosis resulting from changes in diagnostic criteria is that the overall clinical course of MS appears to improve. The average MS patient today has milder disease compared with those diagnosed using the older criteria. However, this may have resulted in lead-time bias, in which earlier diagnosis appears to improve the disease course as patients take longer

to accumulate neurological disability. This would lower the ARR by providing an observed time with fewer relapses relative to the older definitions, such as the Poser criteria. This lead-time bias would also lead to the appearance of less virulent disease, at least in the short term. Whether this is occurring is difficult to separate from potential confounders such as improved availability and efficacy of therapies.

Likewise, a portion of patients with clinically isolated syndrome (CIS) who are at high risk of developing MS will have a diagnosis of MS under the new criteria. The reassignment of patients who are at a higher risk within the CIS group to those having milder disease in the MS group results in the improved overall prognosis of both groups, an effect known as the Will Rogers phenomenon.8 This has been demonstrated since the first iteration of the McDonald criteria by a retrospective analysis showing that 50% of patients with CIS would progress into definite MS within a year by the 2001 McDonald criteria compared with only 20% under the previously used Poser criteria.9 Currently, there is no evidence as to whether the 2017 revision will select patients with less active disease. Recently, a retrospective analvsis on reaching the diagnosis of MS under the latest revision of the McDonald criteria compared with the 2010 revision, concluded that the 2017 revision expedites diagnosis without compromising diagnostic accuracy. 10

# Availability of treatment

The increasing availability of treatment for RRMS has also affected patient enrollment in trials. With multiple DMTs available, there are motivations for high-risk patients not to enroll in a clinical trial as these patients may opt to select a proven therapy rather than enroll in a trial and risk receiving a placebo or a medication with unknown safety or efficacy profiles. Likewise, patients currently involved in trials who experience a relapse may be rescued and removed from the trial to start a proven therapy, thus leaving those whose disease activity is relatively well-controlled during the trial period to complete the study. There are suggestions that over time, withdrawals from therapy have been increasing and even mandated on the occurrence of relapses or progression, thus reducing the opportunity to have subsequent outcome events reducing the ARR and other event rates over the course of the study.

On the other spectrum, patients who are treatment nonresponders may be interested in seeking new options by enrolling in a trial. However, some trials exclude enrollment of patients with extensive exposure to prior DMTs.<sup>11,12</sup>

#### Additional considerations

Other factors have been proposed to account for the changing MS trial population. For example, increasing physician and patient awareness of the disease and accessibility to MRI scanning has led to earlier diagnosis and initiation of treatment. 13,14 Improved standards of care for MS and accessibility to DMTs have resulted in better overall management and thus, lower disease activity in the overall MS population. Modern-day trials allow patients to enroll closer to their time of diagnosis by reducing the requirement of the number of relapses prior to screening. Whereas most early trials required at least two relapses over the course of 2 years, many contemporary trials allow for enrollment with just one relapse over the past year. In addition, recent trials use more stringent and objective definitions of relapse requiring evaluation by a neurologist and measurable increases in the Expanded Disability Status Scale (EDSS) functional system subscores. 15 Together, these changes further substantiate the shift of the MS trial population towards milder disease activity.

# *Implications*

The changing MS trial population complicates the comparison of results from different studies, especially those from different times. 16 Assessing whether one DMT is superior to another is challenging without a head-to-head comparison of the two agents in a randomized controlled trial, yet it is impractical to conduct head-to-head studies of all available agents. As trial patients present with milder disease, it becomes more difficult to generate on-study changes with traditional designs, and therefore larger group sizes and longer study duration are required. At the same time, it is no longer practical to design studies based on assumptions drawn from historic trials. Studies that are designed and statistically powered according to findings from earlier trials may overestimate disease activity and underestimate the number of patients enrolled to be able to detect treatment differences. There is a need for more efficient designs and sensitive outcome measurements.

# Trial designs

The rapid expansion of treatment options for RRMS has led to changes in clinical trial designs over time. The era of placebo-controlled trials to establish new DMTs has ended due to the ethical concerns when existing therapies are available, and new agents are now assessed through comparative studies using established DMTs. The increased costs of large trials and high demand for participants have spurred the need for more efficient designs, while increased therapeutic options calls for pragmatic studies to determine the real-world applicability. The strength and limitations of MS trial designs are summarized in Table 1.

# The ethics of placebo in RRMS trials

New treatment options for RRMS have led to the decline of placebo-controlled trials due to the ethics and marketing concerns of performing such studies when multiple effective therapies exist. The 1990s saw the approval of the first MS therapies, including interferon  $\beta$  preparations and glatiramer acetate. The 1971 These trials relied on the use of a placebo group to demonstrate the efficacy of the investigated agent. Subsequent pivotal phase III trials of new therapies incorporated a placebo group until the approval of alemtuzumab, which in 2014 became the first US-approved agent without comparison with a placebo. 20,21 However, the ethics of using a placebo in MS trials have been in discussion since much earlier.

Following the availability of the first injectable DMTs, the National MS Society organized an international task force in 2000 to discuss and publish proceedings on the ethical use of placebos in MS trials, and its recommendations were further revised in 2008.<sup>22,23</sup> The group concluded that participation in placebo trials is still ethical for patients who decline or fail available agents or when established therapy is not available at the time, such as in PMS. In countries without access to any DMTs, placebos may also remain ethical, but active comparator trials still seem more appropriate. Some countries, such as Brazil, will not allow placebo-controlled trials if the trial is not conducted in the country originating the trial. In all cases, appropriate informed consent and information on available therapies must be provided.

Controversies exist on whether placebo trials are justified, and despite recommendations restricting the use of placebo in RRMS trials, subsequent

**Table 1.** Advantages and disadvantages of MS trial designs.

Trial design	Advantages	Disadvantages	
Placebo-controlled	Most rigorous test of treatment efficacy; requires fewer participants than active comparator trials	No longer ethical in RRMS trials due to availability of proven therapies	
Active comparator	Alternative to placebo-controlled trials and still capable of detecting treatment effect	Requires increased sample size to detect significant treatment differences	
Combination	Potential to increase efficacy by combining therapies	Increased costs and side effects from added treatment; potentially antagonistic interactions; increased design complexity	
Adaptive	Flexible and efficient designs to reduce sample size, exposure to harmful or ineffective treatment, and trial duration	Requires detailed planning and review of interim data as well as sensitive short-term outcome measures	
Pragmatic	Real life evaluation of treatment effectiveness; high external validity; results are more likely to inform practice	Greater design challenges due to heterogeneity of treatment effect; larger sample size and duration needed to determine effectiveness	
MS, multiple sclerosis; RRMS, relapsing-remitting MS.			

phase III trials of fingolimod, teriflunomide, and dimethyl fumarate had approximately 1500 patients exposed to placebo treatment.<sup>24</sup> Many researchers and organizations have taken a middle ground with regard to using a placebo when proven therapies are available.<sup>25</sup> The continued use of a placebo is backed by claims of the changing natural history of MS and indication that data from previous placebo groups are no longer valid.<sup>26</sup> However, as growing evidence surmounts to the detrimental effects of delaying initiation of DMT for RRMS, the use of a placebo in RRMS trials have ended.<sup>27</sup>

# Active comparator designs

The past decade saw a gradual replacement of placebo designs by active comparator trials. The pivotal phase III trial of peginterferon β-1a marked the last placebo-controlled trial of a therapeutic agent for RRMS.<sup>28</sup> Subsequent published phase III trials of daclizumab and ocrelizumab, as well as the recently concluded phase III trial for ozanimod, have been conducted with active comparator arms.<sup>29,30</sup> These trials have shown that active comparator studies are feasible to detect both the efficacy and superiority of new agents. On the other hand, the use of an active comparator arm instead of a placebo reduces on-study changes between both groups, resulting in more patients

needed to observe a significant and meaningful treatment effect. Modern-day trials have reflected this change. Whereas early trials typically enrolled around 200 patients, modern trials have frequently enrolled more than 1000 patients. The increasing costs of conducting large-scale trials have put a strain on existing resources, and there is a need for more efficient designs.

## Combination trials

Combination trials seek to answer the question of whether multiple therapies administered together have greater efficacy than either alone. Studies are designed on the premise of using different agents with complementary or synergistic mechanisms of action.31 In this regard, the most rigorous study to date examined the effects of glatiramer acetate and interferon B-1a administered together, which did not show a benefit in using both drugs over the more effective agent, glatiramer acetate. 32,33 Other studies combining DMTs with hormonal therapy<sup>34</sup> or statins<sup>35</sup> have shown mixed results. Significant drawbacks of employing combination trials include the potential antagonistic mechanisms of action among study agents, compounded costs and side effects, and increased size and complexity of the trial design.36 While these reasons have deterred

combination trials in MS, there may be a role in the future as we progress in understanding the disease pathophysiology and developing more therapeutic options.<sup>37</sup>

# Adaptive designs

Adaptive designs offer flexible and efficient trial methods that allow for preplanned modification of the study protocol after an interim analysis. This allows for flexibility in cost and trial duration. While adaptive designs have seen use in other fields of medicine, current experience with adaptive designs in MS is just emerging. The phase II trial of siponimod for PMS was the first and only published MS trial to date that employed an adaptive design to evaluate the safety and tolerability of the study agent and to determine its dose-efficacy response curve.38 The study randomized one cohort of patients to receive 10 mg, 2 mg, or 0.5 mg of the study agent or placebo and evaluated the outcome of combined unique active MRI lesions at month 3. An interim analysis was performed, and based on an estimation of which doses would be needed for the optimum dose-response characterization, a second cohort of patients were randomized to receive 1.25 mg and 0.25 mg of the study drug. Evidence for the further efficacy of siponimod was recently published in a double-blind randomized placebo phase III trial.<sup>39</sup> The design of adaptive trials demands meticulous planning and a detailed interpretation of the interim results and navigation of a complex review process. It also requires a sensitive short-term outcome measure on which to base interim decisions and long-term outcomes to assess treatment efficacy. Adaptive designs for MS are nascent and their prevalence remains to be seen.

# Pragmatic trials

As therapeutic options for MS broaden, there is a growing need for evidence demonstrating applicability of clinical trial results to routine clinical practice. In contrast with explanatory trials, which measure efficacy under controlled settings, pragmatic trials are designed to evaluate treatment effectiveness in the real-world setting, producing results that are more generalizable and applicable to clinical practice. Despite the distinction between explanatory and pragmatic trials, actual trials fall on a spectrum between the two categories as characterized the Pragmatic–Explanatory Continuum Index Summary (PRECIS) tool and its subsequent revision (PRECIS-2). 41,42 The tool

helps trialists to ensure their design decisions match the trial's intended purpose.

Just as explanatory trials face the challenge of requiring a large sample size with enough followup time to detect significant treatment effects, the issue is even greater in pragmatic trials when selection criteria and interventions are less controlled. A purely pragmatic trial in MS is almost unattainable and will almost certainly include components towards the explanatory end of the spectrum. As adeptly summarized by Ford and Norrie in a publication of pragmatic trials in the New England Journal of Medicine, 'a pragmatic approach to pragmatism would be to adopt the features of pragmatic trials whenever feasible and sensible and when such features do not compromise trial quality and the ability to answer the clinical question of interest.'43

#### Traditional and new outcome measures

Relevant and sensitive outcome measures are key to the assessment of a new drug's efficacy. Traditionally, phase II trials assess the short-term effects on MRI outcomes, while phase III designs focus on more overt clinical outcomes. While this model has standardized the development of the currently available DMTs, the outcome measures used are imperfect, largely due to the heterogeneity of the disease. The paradigm of the MRI in phase II may be limited to the mechanism of action of the DMTs; most, if not all, have addressed inflammation as the target. New drugs may require different measures either on MRI or other biomarkers. Variations in disease presentation across the population and even within an individual over time have led to challenges in the development of methods to assess treatment efficacy. This is further complicated by the need to capture changes within a short clinical trial duration for a disease with a prolonged clinical course. Traditional clinical and imaging outcome measures continue to show utility in modern trials while new measures in advanced imaging outcomes, biomarkers and composite outcomes have diversified options for choosing study endpoints. The advantages and disadvantages of outcomes measures in MS trials are summarized in Table 2.

# Clinical outcome measures

Clinical outcomes play an important role as the primary endpoint of pivotal phase III trials.

**Table 2.** Advantages and disadvantages of outcomes measurements in MS trials.

Outcome measure	Advantages	Disadvantages
EDSS	Established and universally accepted clinical outcome measurement; familiar to clinicians and regulators	Inter-rater variability; heavy emphasis on walking; exclusion of cognitive impairment; nonlinear scale
MSFC	High reliability, validity, and sensitivity; versatile in evaluating various levels of disability	z-scores are abstract and difficult to interpret
Patient-reported outcomes	Patient-centered assessment of disease or treatment impact	Inherently subjective; limited use in disease-modifying therapy efficacy trials
MRI lesions	Objective and quantifiable; High sensitivity in detecting subclinical disease activity; particularly useful as primary outcome measures in phase II trials	Lesions do not reflect degree of clinical disability
Brain atrophy	Correlations with disability and cognitive impairment; measurement of neurodegeneration	Requires longer time to detect atrophy changes; multiple variables may confound measurements
ОСТ	Fast, inexpensive, noninvasive technique; association with neurodegeneration	Limited evaluation of central nervous system function and disease burden
Fluid biomarkers	Provides insight into disease pathophysiology; detects ongoing disease process; easy to obtain from serum	No established biomarkers; current biomarkers lack specificity
Combined outcomes (NEDA-3 and NEDA-4)	Better predictive value for disability progression than individual measures can identify treatment response	Cannot measure specific outcomes; no standardized set of components

EDSS, Expanded Disability Status Scale; MRI, magnetic resonance imaging; MS, multiple sclerosis; MSFC, multiple sclerosis functional composite; NEDA-3, no evidence of disease activity including absence of new lesions, confirmed EDSS worsening and relapses; NEDA-4, as NEDA-3 in addition to absence of significant brain volume changes; OCT, optical coherence tomography.

Assessments of relapse and changes in the EDSS score have been the oldest and most widely used clinical outcome measures. They each capture distinct pathophysiologic processes of the disease. Whereas relapse rates measure inflammatory activity, changes in the EDSS reflect accumulation of clinical disability from inflammatory or neurodegenerative disease processes. The EDSS is universally familiar to MS clinicians and accepted by regulators, 44,45 but it is marked by shortcomings in its variability between examiners, heavy emphasis on walking, limited impact of cognitive impairment and nonlinearity. The imperfections of the EDSS have led to the development of the MS functional composite (MSFC), 46 which

covers three major MS domains, can be assessed quickly and reliably, and gives a *z*-score on a continuous scale. However, *z*-scores are abstract and difficult to interpret clinically, which has limited the widespread use of the MSFC. Despite its imperfections, the EDSS currently remains the most accepted clinical outcome measure and will likely remain so for the near future. Recently, electronic scoring of the EDSS has been developed and has shown an increased consistency of EDSS scoring,<sup>47</sup> and it has seen use in the phase III trials of ocrelizumab<sup>30</sup> and siponimod.<sup>39</sup>

Over the past decade, patient-reported outcome (PRO) measures have increasingly been used as

complements to objective clinical measures and provide important information on subjective outcomes, such as evaluations of subjective symptoms, satisfaction with treatment, and quality of life.<sup>48–50</sup> They offer an additional facet in the assessment of treatment risks and benefits. Nevertheless, the subjectivity of PROs typically limits their use to a secondary outcome measure for DMT investigations.<sup>51</sup> Their use as primary outcomes for symptomatic therapies may be appropriate.

# Neuroimaging outcome measures

Neuroimaging outcome measures provide additional information to supplement clinical measurements of treatment efficacy. Most commonly used are MRI measurements of lesion load or atrophy, which provide objective and highly sensitive measurements of current and past disease activity. Inflammatory activity can be quantified as focal lesions on T2 MRI sequences or T1 gadolinium-enhancing lesions and have shown a correlation with relapse rate. 52,53 This important metric has allowed for rapid development of DMTs for RRMS due to its use as the primary outcome in short-term phase II trials. The ability to detect reductions in lesions on MRI associated with treatment allows for quick progress into phase III trials.54 The accrual of disability as reflected by neurodegeneration is measured by brain atrophy. This has been commonly assessed either as total brain parenchymal volume or fractional volume, defined as the ratio of brain parenchymal volume to total volume within the brain surface contour.

MRI outcome measurements have undergone refinement as the understanding of the disease process continues to develop. Composite imaging outcomes of combined unique active lesions have seen use in a growing number of trials to assess total new or enlarging T2 or gadolinium-enhancing lesions. Regional atrophy is increasingly explored as a more sensitive measurement of changes reflective of associated deficits, and areas of the thalamus and cerebellum may be more sensitive to changes.<sup>55,56</sup> Gray matter atrophy has been shown to have more clinical correlation than white matter atrophy and is more sensitive to changes.<sup>57</sup>

Optical coherence tomography (OCT) is a noninvasive method to image the retina and characterize pathology of the optic nerve.<sup>58</sup> By visualizing

the unmyelinated axons of the retinal nerve fiber layer (RNFL), OCT can directly assess axonal injury. In MS patients, the RNFL is thinner than in healthy people, regardless of a previous history of optic neuritis, supporting its role as a unique *in vivo* model of neurodegeneration. <sup>59–62</sup> OCT has been used as an outcome measure in a few clinical trials in PMS without evidence of neuroprotective effects. <sup>63–66</sup> Despite its current lack of prevalence in clinical trials, OCT as a fast, inexpensive, and noninvasive approach to evaluate neurodegeneration, continues to prompt further investigation into refining the method.

Advancements in imaging techniques have led to the development of new outcome measures. Magnetization transfer ratio (MTR) is an MRI technique used to assess tissue integrity by correlating it with axonal loss in MS lesions. Whole brain MTR has been used in several trials, <sup>28,63,64,67,68</sup> while regional MTR has been used in the phase III trial of laquinimod for RRMS. <sup>69</sup> While imaging outcome measurements are not likely to replace clinical endpoints in trials, they provide important information that supplements data on the efficacy of novel agents.

## Fluid biomarkers

Clinical and radiologic outcome measures have served as the cornerstone of assessing the therapeutic efficacy of drugs and disease activity; however, they provide limited assessment of the pathophysiological disease process, such as inflammation, axonal injury, and demyelination. Early search for biomarkers in the spinal fluid of MS patients hoped to identify outcomes that could be used in trials, but these efforts have not yielded promising results. In recent years, several biomarkers for MS have shown potential for diagnosis, prognosis, monitoring disease activity and treatment response.<sup>70</sup> Although currently there are no established biomarkers in use in clinical trials, among the most promising is neurofilament light chain (NFL).71 NFL is a subunit of neurofilaments that compose the axonal and dendritic cytoskeleton. It is released during axonal injury, making it a marker of axonal damage.<sup>72</sup> In addition to its presence in the cerebrospinal fluid, NFL can also be detected in the serum due to using the ultrasensitive single molecule array (Simoa) technology.<sup>73,74</sup> Serum NFL has been shown to correlate with relapses, EDSS scores, MRI lesions, and brain and spinal cord atrophy,75-78 and its levels have shown

a decrease in patients treated with DMTs.<sup>79–82</sup> Its downside is the lack of specificity, and levels may be influenced by many comorbidities or conditions. With an increased availability and understanding of NFL and disease activity, NFL may become the first fluid biomarker to be used to monitor therapeutic effects in future randomized clinical trials and we may eventually see its use in the clinical setting.

# Combined outcome measures

Other new outcome measures have integrated domains to achieve a more holistic evaluation.83 The concept of disease activity-free status in MS was first explored in the natalizumab phase III trial,84 and it was later defined as 'no evidence of disease activity' (NEDA) to reflect the absence of relapses and no progression of disability or new MRI activity. NEDA has been used in some phase II<sup>85,86</sup> and phase III trials.<sup>20,21,32</sup> The combination of outcome measures increased the predictive value of disability progression compared with individual clinical or MRI metrics.87 The drawback of using NEDA lies in the inability to assess its effects on specific outcomes, since clinical and radiological activity are all combined into one assessment that is heavily dependent on timing. Currently, NEDA is still undergoing refinement as it remains unclear which optimal components to include and how frequently they should be assessed. In addition, brain volume as a marker of neurodegeneration has been proposed to be included in a revised measure referred to as NEDA-4.88 As the concept of NEDA continues to evolve, it becomes important to establish standardized definitions to facilitate universal interpretation and demonstrate its validity and reliability across studies.

# Clinical trials for progressive MS

While substantial progress has been made in the advancement of therapeutic options for RRMS, progress for PMS remains limited and treatments are much needed, and only recently has a DMT been approved for PMS.<sup>89</sup> The challenges in advancing therapeutics in PMS arise from a limited understanding of the pathogenesis of PMS and difficulty defining and assessing the disease course, as well as apparent confounding with the aging process. This contrasts with the established trial methodology for RRMS in which changes in focal MRI lesions in phase II trials allow for the accurate prediction of clinical reductions in relapse

rate in phase III trials. Trials for PMS do not have established enrollment criteria owing to the lack of a uniform clinical definition of the disease. Ourrently, there are no sensitive markers of disease activity that can accurately capture the varying degrees of inflammatory activity in the setting of neurodegeneration. Due to the process of neurodegeneration and compartmentalized inflammation playing a key role in PMS, drugs with predominantly systemic anti-inflammatory activity are less likely to be effective, especially in older patients where inflammatory activity has declined.

There is a heightened focus on clinical trials for PMS. Recently, The European Committee for Treatment and Research in MS (ECTRIMS) in association with the International Progressive MS Alliance held a workshop to address strategies for the advancement of trial design in PMS.91,92 The workshop addressed multiple aspects of trials for PMS and proceedings were published as a series of review articles in a themed issue of the Multiple Sclerosis Yournal. The group recognized the growing importance of people with PMS in providing feedback and marketing and even in the design of trials.93 Patients now undertake a larger role than just participating in volunteer trials. They are involved in providing feedback in developing more clinically meaningful outcome measurements, advocating financial support, and serving as representatives in the trial's development committees.

The group recognized the pathophysiology of PMS as playing a key role in the trial design and therapy selection. The pathophysiology of PMS differs from RRMS in which systemic inflammation progresses to compartmentalized inflammation and neurodegeneration. The dynamic changes in pathophysiology over the course of the disease highlights the importance of identifying the disease stage and timing of intervention. Drugs with anti-inflammatory properties will have more benefit in patients with active inflammation and are less effective in patients who have reached the late stages of neurodegeneration.94 Enrollment of patients who have more active disease progression will likely have a greater sensitivity in detecting the treatments measuring progression than patients who do not progress. However, too narrow inclusion criteria will compromise the external validity and generalizability of results. More research to understand the pathophysiology and population characteristics of PMS are underway.

Due to a current lack of early and sensitive measurements of disease progression in PMS, an emphasis was placed on outcome measures. Clinical outcome measures such as the EDSS continue to serve as the established benchmark in PMS trials, but the shortcomings of EDSS limit its use alone as a sufficient outcome because of its near exclusive reliance in its upper ranges on walking and ignoring important functions such as cognitive and arm functions. Other measures have begun to surface that assess the vital clinical components of PMS. Together, these outcome measures address multiple facets of the clinical experience and are shaping trials to include a reworking and integration of new metrics such as composite measurements, cognition, and PROs. 95 The lack of a sensitive phase II trial outcome measure poses a limit to the timely advancement of potential therapeutic agents beyond their proof-of-concept phase. This contrasts with phase II trials for RRMS that employ focal MRI lesions to detect short-term changes. Currently, whole brain atrophy is the most accepted phase II outcome measurement in PMS.96 However, there are notable limitations including slowness to change, inter-patient variability, and unidimensional measurement. Other biomarkers in development and testing, such as NFL, have begun to show promise.97

Despite challenges in PMS trials, there is progress in recent trials that have shown a positive effect in PMS including simvastatin, <sup>98</sup> biotin, <sup>99</sup> ocrelizumab, <sup>89</sup> and siponimod. <sup>39</sup> Potential evolution in the trials in PMS will focus on identifying agents targeting neuroprotection and remyelination, carefully identifying a study population with regards to baseline inflammatory activity, choosing accurate and sensitive outcome measures in both phase II and III trials, and exploring the potential of adaptive designs.

# Conclusion

The past 25 years have witnessed substantial developments in the treatment of MS. Advancements in therapeutic agents are a direct result of clinical trials that demonstrated their efficacy. Just as the disease course has been redefined with the advent of DMTs, so have characteristics of trials that continue to test new agents. The present-day MS trial population no longer shares the same baseline characteristics as historical groups and is distinguished by earlier

diagnosis and milder disease presentation due to increased availability of treatment. This change is present even in placebo groups, thus complicating the comparison of data across trials. At the same time, active comparator designs have replaced placebo-controlled trials for RRMS, as the latter is no longer ethical in an era of proven therapeutic options. The head-to-head comparison of two agents has increased the required trial sample size and duration to be able to detect significant on-study differences. This has motivated consideration of more efficient study designs such as adaptive designs that have the potential to seamlessly transition from phase II to phase III trials, thus reducing the cost and expediting the trial process.

At the same time, new options for clinical and neuroimaging outcome measurements are beginning to see use in trials to capture different aspects of the disease process. While traditional measures of relapse (e.g. changes in EDSS and focal MRI lesions) continue to remain the standard of assessment of disease activity, new parameters such as brain volume, PROs, MTR, OCT, NFL have contributed new dimensions to capturing disease progression.

Amidst the successful development of DMTs for RRMS, increasing focus has turned towards clinical trials for PMS where current therapies are limited and much needed. Experience with PMS trials has identified challenges in study design inherent in the nebulous characterization of the disease course and lack of sensitive measurements of disease progression. Unlike in RRMS where inflammatory activity can be assessed by relapses and lesion changes on MRI, there is no sensitive equivalent in measuring neurodegeneration and compartmentalized inflammation, which constitutes significant mechanisms in the pathophysiology of PMS. Nevertheless, recent research has shown positive results in multiple agents for the treatment of PMS, and outcome measures such as brain atrophy is an acceptable metric of neurodegeneration. Various agents in remyelination and neuroprotection are currently being tested, and the potential applications of adaptive designs are being explored.

# **Funding**

This research received no specific grant from any funding agency in the public, commercial, or not-for-profit sectors.

## Conflict of interest statement

The authors declare that there is no conflict of interest.

#### References

- De Gasperis-Brigante CD, Parker JL, O'Connor PW, et al. Reducing clinical trial risk in multiple sclerosis. Mult Scler Relat Disord 2016; 5: 81–88.
- Uitdehaag BM, Barkhof F, Coyle PK, et al.
   The changing face of multiple sclerosis clinical trial populations. Curr Med Res Opin 2011; 27: 1529–1537.
- Montalban X. Review of methodological issues of clinical trials in multiple sclerosis. J Neurol Sci 2011; 311(Suppl. 1): S35–S42.
- 4. Inusah S, Sormani MP, Cofield SS, *et al.*Assessing changes in relapse rates in multiple sclerosis. *Mult Scler* 2010; 16: 1414–1421.
- Ntranos A and Lublin F. Diagnostic criteria, classification and treatment goals in multiple sclerosis: the chronicles of time and space. *Curr Neurol Neurosci Rep* 2016; 16: 90.
- Brownlee WJ, Hardy TA, Fazekas F, et al. Diagnosis of multiple sclerosis: progress and challenges. Lancet 2017; 389: 1336–1346.
- Thompson AJ, Banwell BL, Barkhof F, et al. Diagnosis of multiple sclerosis: 2017 revisions of the McDonald criteria. Lancet Neurol 2018; 17: 162–173.
- 8. Sormani MP, Tintore M, Rovaris M, *et al.* Will Rogers phenomenon in multiple sclerosis. *Ann Neurol* 2008; 64: 428–433.
- 9. Dalton CM, Brex PA, Miszkiel KA, et al. Application of the new McDonald criteria to patients with clinically isolated syndromes suggestive of multiple sclerosis. *Ann Neurol* 2002; 52: 47–53.
- Beesley R, Anderson V, Harding KE, et al. Impact of the 2017 revisions to McDonald criteria on the diagnosis of multiple sclerosis. Mult Scler 2018: 24: 1786–1787.
- O'Connor P, Filippi M, Arnason B, et al. 250 microg or 500 microg interferon beta-1b versus 20 mg glatiramer acetate in relapsing-remitting multiple sclerosis: a prospective, randomised, multicentre study. *Lancet Neurol* 2009; 8: 889–897.
- 12. Mikol DD, Barkhof F, Chang P, *et al*.

  Comparison of subcutaneous interferon beta-1a with glatiramer acetate in patients with relapsing

- multiple sclerosis (the REbif vs Glatiramer Acetate in Relapsing MS Disease [REGARD] study): a multicentre, randomised, parallel, open-label trial. *Lancet Neurol* 2008; 7: 903–914.
- 13. Goodin DS and Bates D. Treatment of early multiple sclerosis: the value of treatment initiation after a first clinical episode. *Mult Scler* 2009; 15: 1175–1182.
- Miller JR. The importance of early diagnosis of multiple sclerosis. J Manag Care Pharm 2004; 10: S4–S11.
- 15. Tur C, Moccia M, Barkhof F, *et al.* Assessing treatment outcomes in multiple sclerosis trials and in the clinical setting. *Nat Rev Neurol* 2018; 14: 75–93.
- Mitsikostas DD and Goodin DS. Comparing the efficacy of disease-modifying therapies in multiple sclerosis. *Mult Scler Relat Disord* 2017; 18: 109–116.
- 17. Interferon beta-1b is effective in relapsing-remitting multiple sclerosis. I. Clinical results of a multicenter, randomized, double-blind, placebo-controlled trial. The IFNB Multiple Sclerosis Study Group. *Neurology* 1993; 43: 655–661.
- Jacobs LD, Cookfair DL, Rudick RA, et al.
   Intramuscular interferon beta-1a for disease progression in relapsing multiple sclerosis.
   The Multiple Sclerosis Collaborative Research Group (MSCRG). Ann Neurol 1996; 39: 285–294.
- 19. Johnson KP, Brooks BR, Cohen JA, et al. Copolymer 1 reduces relapse rate and improves disability in relapsing-remitting multiple sclerosis: results of a phase III multicenter, double-blind placebo-controlled trial. The Copolymer 1 Multiple Sclerosis Study Group. Neurology 1995; 45: 1268–1276.
- 20. Cohen JA, Coles AJ, Arnold DL, *et al.*Alemtuzumab versus interferon beta 1a as first-line treatment for patients with relapsing-remitting multiple sclerosis: a randomised controlled phase 3 trial. *Lancet* 2012; 380: 1819–1828.
- 21. Coles AJ, Twyman CL, Arnold DL, *et al.* Alemtuzumab for patients with relapsing multiple sclerosis after disease-modifying therapy: a randomised controlled phase 3 trial. *Lancet* 2012; 380: 1829–1839.
- 22. Lublin FD and Reingold SC. Placebocontrolled clinical trials in multiple sclerosis: ethical considerations. National Multiple

- Sclerosis Society (USA) Task Force on Placebo-Controlled Clinical Trials in MS. *Ann Neurol* 2001; 49: 677–681.
- 23. Polman CH, Reingold SC, Barkhof F, et al. Ethics of placebo-controlled clinical trials in multiple sclerosis: a reassessment. *Neurology* 2008; 70: 1134–1140.
- 24. Solomon AJ and Bernat JL. A review of the ethics of the use of placebo in clinical trials for relapsing-remitting multiple sclerosis therapeutics. *Mult Scler Relat Disord* 2016; 7: 109–112.
- 25. Emanuel EJ and Miller FG. The ethics of placebo-controlled trials–a middle ground. *N Engl J Med* 2001; 345: 915–919.
- Zajicek J. Future placebo-controlled trials of disease modifying therapy in relapsing multiple sclerosis would be unethical: No. *Mult Scler* 2014; 20: 1165–1166.
- 27. Hawkins S. Future placebo-controlled trials of disease modifying therapy in relapsing multiple sclerosis would be unethical: Yes. *Mult Scler* 2014; 20: 1167–1168.
- Calabresi PA, Kieseier BC, Arnold DL, et al.; ADVANCE Study Investigators. Pegylated interferon beta-1a for relapsing-remitting multiple sclerosis (ADVANCE): a randomised, phase 3, double-blind study. Lancet Neurol 2014; 13: 657–665.
- Kappos L, Wiendl H, Selmaj K, et al.
   Daclizumab HYP versus interferon beta-1a in relapsing multiple sclerosis. N Engl J Med 2015;
   373: 1418–1428.
- Hauser SL, Bar-Or A, Comi G, et al.
   Ocrelizumab versus interferon beta-1a in relapsing multiple sclerosis. N Engl J Med 2017; 376; 221–234.
- Milo R and Panitch H. Combination therapy in multiple sclerosis. J Neuroimmunol 2011; 231: 23–31.
- 32. Lublin FD, Cofield SS, Cutter GR, *et al.*Randomized study combining interferon and glatiramer acetate in multiple sclerosis. *Ann Neurol* 2013; 73: 327–340.
- Lublin FD, Cofield SS, Cutter GR, et al.
   Long-term follow-up of a randomized study of combination interferon and glatiramer acetate in multiple sclerosis: efficacy and safety results up to 7 years. Mult Scler Relat Disord 2017; 18: 95–102.
- Voskuhl RR, Wang H, Wu TC, et al. Estriol combined with glatiramer acetate for women

- with relapsing-remitting multiple sclerosis: a randomised, placebo-controlled, phase 2 trial. *Lancet Neurol* 2016; 15: 35–46.
- Bhardwaj S, Coleman CI and Sobieraj DM. Efficacy of statins in combination with interferon therapy in multiple sclerosis: a metaanalysis. Am J Health Syst Pharm 2012; 69: 1494–1499.
- Conway D and Cohen JA. Combination therapy in multiple sclerosis. *Lancet Neurol* 2010; 9: 299–308.
- 37. Avasarala J. It's time for combination therapies: in multiple sclerosis. *Innov Clin Neurosci* 2017; 14: 28–30.
- 38. Selmaj K, Li DK, Hartung HP, *et al.*Siponimod for patients with relapsing-remitting multiple sclerosis (BOLD): an adaptive, doseranging, randomised, phase 2 study. *Lancet Neurol* 2013; 12: 756–767.
- 39. Kappos L, Bar-Or A, Cree BAC, *et al.* Siponimod versus placebo in secondary progressive multiple sclerosis (EXPAND): a double-blind, randomised, phase 3 study. *Lancet* 2018; 391: 1263–1273.
- Schwartz D and Lellouch J. Explanatory and pragmatic attitudes in therapeutical trials. *J* Chronic Dis 1967; 20: 637–648.
- 41. Thorpe KE, Zwarenstein M, Oxman AD, et al. A pragmatic-explanatory continuum indicator summary (PRECIS): a tool to help trial designers. J Clin Epidemiol 2009; 62: 464–475.
- 42. Loudon K, Treweek S, Sullivan F, *et al.* The PRECIS-2 tool: designing trials that are fit for purpose. *BM*7 2015; 350: h2147.
- Ford I and Norrie J. Pragmatic trials. N Engl J Med 2016; 375: 454–463.
- 44. Cohen JA, Reingold SC, Polman CH, *et al.*Disability outcome measures in multiple sclerosis clinical trials: current status and future prospects. *Lancet Neurol* 2012; 11: 467–476.
- 45. Hyland M and Rudick RA. Challenges to clinical trials in multiple sclerosis: outcome measures in the era of disease-modifying drugs. *Curr Opin Neurol* 2011; 24: 255–261.
- 46. Cutter GR, Baier ML, Rudick RA, *et al.*Development of a multiple sclerosis functional composite as a clinical trial outcome measure. *Brain* 1999; 122: 871–882.
- 47. D'Souza M, Yaldizli O, John R, *et al.*Neurostatus e-Scoring improves consistency of Expanded Disability Status Scale assessments:

- a proof of concept study. *Mult Scler* 2017; 23: 597–603.
- Lublin F, Miller DH, Freedman MS, et al.; INFORMS study investigators. Oral fingolimod in primary progressive multiple sclerosis (INFORMS): a phase 3, randomised, doubleblind, placebo-controlled trial. *Lancet* 2016; 387: 1075–1084.
- 49. Andersen O, Elovaara I, Farkkila M, et al. Multicentre, randomised, double blind, placebo controlled, phase III study of weekly, low dose, subcutaneous interferon beta-1a in secondary progressive multiple sclerosis. *J Neurol Neurosurg Psychiatry* 2004; 75: 706–710.
- Zajicek J, Ball S, Wright D, et al. Effect of dronabinol on progression in progressive multiple sclerosis (CUPID): a randomised, placebo-controlled trial. Lancet Neurol 2013; 12: 857–865.
- 51. Speight J and Barendse SM. FDA guidance on patient-reported outcomes. *BMJ* 2010; 340: c2921.
- Tintore M, Rovira A, Rio J, et al. Defining high, medium and low impact prognostic factors for developing multiple sclerosis. Brain 2015; 138: 1863–1874.
- 53. Sormani MP and Bruzzi P. MRI lesions as a surrogate for relapses in multiple sclerosis: a meta-analysis of randomised trials. *Lancet Neurol* 2013; 12: 669–676.
- 54. Stone LA, Frank JA, Albert PS, et al. Characterization of MRI response to treatment with interferon beta-1b: contrastenhancing MRI lesion frequency as a primary outcome measure. *Neurology* 1997; 49: 862–869.
- 55. Cocozza S, Petracca M, Mormina E, et al. Cerebellar lobule atrophy and disability in progressive MS. J Neurol Neurosurg Psychiatry 2017; 88: 1065–1072.
- Bergsland N, Zivadinov R, Dwyer MG, et al. Localized atrophy of the thalamus and slowed cognitive processing speed in MS patients. Mult Scler 2016; 22: 1327–1336.
- Fisniku LK, Chard DT, Jackson JS, et al. Gray matter atrophy is related to long-term disability in multiple sclerosis. Ann Neurol 2008; 64: 247–254.
- 58. Drexler W and Fujimoto JG. State-of-the-art retinal optical coherence tomography. *Prog Retin Eye Res* 2008; 27: 45–88.

- Saidha S, Al-Louzi O, Ratchford JN, et al.
   Optical coherence tomography reflects brain atrophy in multiple sclerosis: a four-year study.
   Ann Neurol 2015; 78: 801–813.
- Petzold A, de Boer JF, Schippling S, et al.
   Optical coherence tomography in multiple sclerosis: a systematic review and meta-analysis.
   Lancet Neurol 2010; 9: 921–932.
- 61. Fisher JB, Jacobs DA, Markowitz CE, *et al.* Relation of visual function to retinal nerve fiber layer thickness in multiple sclerosis. *Ophthalmology* 2006; 113: 324–332.
- 62. Alonso R, Gonzalez-Moron D and Garcea O. Optical coherence tomography as a biomarker of neurodegeneration in multiple sclerosis: a review. *Mult Scler Relat Disord* 2018; 22: 77–82.
- 63. Connick P, Kolappan M, Crawley C, *et al.*Autologous mesenchymal stem cells for the treatment of secondary progressive multiple sclerosis: an open-label phase 2a proof-of-concept study. *Lancet Neurol* 2012; 11: 150–156.
- 64. Fox RJ, Coffey CS, Cudkowicz ME, *et al.*Design, rationale, and baseline characteristics of the randomized double-blind phase II clinical trial of ibudilast in progressive multiple sclerosis. *Contemp Clin Trials* 2016; 50: 166–177.
- 65. Cambron M, Mostert J, Haentjens P, *et al.* Fluoxetine in progressive multiple sclerosis (FLUOX-PMS): study protocol for a randomized controlled trial. *Trials* 2014; 15: 37.
- 66. Rice CM, Marks DI, Ben-Shlomo Y, *et al.*Assessment of bone marrow-derived cellular therapy in progressive multiple sclerosis (ACTiMuS): study protocol for a randomised controlled trial. *Trials* 2015; 16: 463.
- 67. Miller DH, Fox RJ, Phillips JT, *et al.* Effects of delayed-release dimethyl fumarate on MRI measures in the phase 3 CONFIRM study. *Neurology* 2015; 84: 1145–1152.
- 68. Romme Christensen J, Ratzer R, Bornsen L, et al. Natalizumab in progressive MS: results of an open-label, phase 2A, proof-of-concept trial. Neurology 2014; 82: 1499–1507.
- Comi G, Jeffery D, Kappos L, et al. Placebocontrolled trial of oral laquinimod for multiple sclerosis. N Engl J Med 2012; 366: 1000–1009.
- Comabella M and Montalban X. Body fluid biomarkers in multiple sclerosis. *Lancet Neurol* 2014; 13: 113–126.
- 71. Lycke J and Zetterberg H. The role of blood and CSF biomarkers in the evaluation of new treatments against multiple sclerosis.

- Expert Rev Clin Immunol 2017; 13: 1143–1153.
- Norgren N, Rosengren L and Stigbrand T. Elevated neurofilament levels in neurological diseases. *Brain Res* 2003; 987: 25–31.
- 73. Rissin DM, Kan CW, Campbell TG, *et al.*Single-molecule enzyme-linked immunosorbent assay detects serum proteins at subfemtomolar concentrations. *Nat Biotechnol* 2010; 28: 595–599.
- 74. Gisslen M, Price RW, Andreasson U, et al. Plasma concentration of the neurofilament light protein (NFL) is a biomarker of CNS injury in HIV infection: a cross-sectional study. EBioMedicine 2016; 3: 135–140.
- Barro C, Benkert P, Disanto G, et al. Serum neurofilament as a predictor of disease worsening and brain and spinal cord atrophy in multiple sclerosis. Brain 2018; 141: 2382-2391.
- Novakova L, Zetterberg H, Sundstrom P, et al. Monitoring disease activity in multiple sclerosis using serum neurofilament light protein. Neurology 2017; 89: 2230–2237.
- 77. Kuhle J, Nourbakhsh B, Grant D, *et al.* Serum neurofilament is associated with progression of brain atrophy and disability in early MS. *Neurology* 2017; 88: 826–831.
- 78. Disanto G, Barro C, Benkert P, *et al.* Serum neurofilament light: a biomarker of neuronal damage in multiple sclerosis. *Ann Neurol* 2017; 81: 857–870.
- Novakova L, Axelsson M, Khademi M, et al.
   Cerebrospinal fluid biomarkers as a measure of disease activity and treatment efficacy in relapsing-remitting multiple sclerosis. J Neurochem 2017; 141: 296–304.
- Kuhle J, Disanto G, Lorscheider J, et al.
   Fingolimod and CSF neurofilament light chain levels in relapsing-remitting multiple sclerosis.

   Neurology 2015; 84: 1639–1643.
- 81. Novakova L, Axelsson M, Khademi M, et al. Cerebrospinal fluid biomarkers of inflammation and degeneration as measures of fingolimod efficacy in multiple sclerosis. *Mult Scler* 2017; 23: 62–71.
- 82. Axelsson M, Malmestrom C, Gunnarsson M, *et al.* Immunosuppressive therapy reduces axonal damage in progressive multiple sclerosis. *Mult Scler* 2014; 20: 43–50.

- van Munster CE and Uitdehaag BM. Outcome measures in clinical trials for multiple sclerosis. CNS Drugs 2017; 31: 217–236.
- 84. Polman CH, O'Connor PW, Havrdova E, *et al.* A randomized, placebo-controlled trial of natalizumab for relapsing multiple sclerosis. *N Engl J Med* 2006; 354: 899–910.
- 85. Burman J, Iacobaeus E, Svenningsson A, et al. Autologous haematopoietic stem cell transplantation for aggressive multiple sclerosis: the Swedish experience. J Neurol Neurosurg Psychiatry 2014; 85: 1116–1121.
- Atkins HL, Bowman M, Allan D, et al.
   Immunoablation and autologous haemopoietic stem-cell transplantation for aggressive multiple sclerosis: a multicentre single-group phase 2 trial. Lancet 2016; 388: 576–585.
- 87. Rotstein DL, Healy BC, Malik MT, *et al.* Evaluation of no evidence of disease activity in a 7-year longitudinal multiple sclerosis cohort. *JAMA Neurol* 2015; 72: 152–158.
- 88. Kappos L, De Stefano N, Freedman MS, *et al.* Inclusion of brain volume loss in a revised measure of 'no evidence of disease activity' (NEDA-4) in relapsing-remitting multiple sclerosis. *Mult Scler* 2016; 22: 1297–1305.
- 89. Montalban X, Hauser SL, Kappos L, *et al.* Ocrelizumab versus placebo in primary progressive multiple sclerosis. *N Engl J Med* 2017; 376: 209–220.
- Lorscheider J, Buzzard K, Jokubaitis V, et al. Defining secondary progressive multiple sclerosis. Brain 2016; 139: 2395–2405.
- Miller DH and Thompson AJ. Advancing trial design in progressive multiple sclerosis. *Mult Scler* 2017; 23: 1571–1572.
- Fox RJ and Chataway J. Advancing trial design in progressive multiple sclerosis. *Mult Scler* 2017; 23: 1573–1578.
- 93. Smith K. The evolving role of people with MS in clinical research-Some progress but more is needed. *Mult Scler* 2017; 23: 1579–1582.
- 94. Lassmann H. Targets of therapy in progressive MS. *Mult Scler* 2017; 23: 1593–1599.
- 95. Ontaneda D, Cohen JA and Amato MP. Clinical outcome measures for progressive MS trials. *Mult Scler* 2017; 23: 1627–1635.
- 96. Ontaneda D and Fox RJ. Imaging as an outcome measure in multiple sclerosis. *Neurotherapeutics* 2017; 14: 24–34.

97. Barro C, Leocani L, Leppert D, *et al.* Fluid biomarker and electrophysiological outcome measures for progressive MS trials. *Mult Scler* 2017; 23: 1600–1613.

98. Chataway J, Schuerer N, Alsanousi A, *et al.*Effect of high-dose simvastatin on brain atrophy and disability in secondary progressive multiple sclerosis (MS-STAT): a randomised,

- placebo-controlled, phase 2 trial. *Lancet* 2014; 383: 2213–2221.
- 99. Tourbah A, Lebrun-Frenay C, Edan G, et al. MD1003 (high-dose biotin) for the treatment of progressive multiple sclerosis: a randomised, double-blind, placebocontrolled study. *Mult Scler* 2016; 22: 1719–1731.

Visit SAGE journals online journals.sagepub.com/ home/tan

**\$**SAGE journals